Author's response to reviews

Title: Spontaneous Corneal Melting in Pregnancy: A case report

Authors:

Sudesh K Arya (aryasudesh@yahoo.co.in)
Archana Malik (drarchanag2002@yahoo.com)
Sonika Gupta (sonika_gupta@yahoo.com)
Hemlata Gupta (hemlatagupta@rediffmail.com)
Sunandan Sood (soods32@yahoo.co.in)

Version: 8 Date: 2 November 2007

Author's response to reviews:

To,
The Editor
JMCR

Sub: Submission of revised version

Sir,

Written and informed consent of the patient was obtained and I have mentioned that at the end of the manuscript. I am resending Figure 2 in a revised format. Please let me know if any other alteration is required.

Archana